Author’s response to reviews

Title: Household food insecurity and symptoms of neurologic disorder in Ethiopia: An observational analysis

Authors:

  Abdulrahman M El-Sayed (ame2145@columbia.edu)
  Craig Hadley (chadley@emory.edu)
  Fasil Tessema (fasil.tessema@ju.edu.et)
  Ayelew Tegegn (ayelewtm@yahoo.com)
  John A Cowan Jr (jacowan@umich.edu)
  Sandro Galea (sgalea@columbia.edu)

Version: 6 Date: 24 December 2010

Author’s response to reviews: see over
December 24, 2010

Dear Dr Norton:

Thank you for the final review of our manuscript, and for accepting it for publication. We are grateful for your and the reviewers’ thorough edits, which we believe have led to an improved manuscript. We have, in this revised version, addressed all reviewer comments and summarize our edits to the MS below.

**Editorial adjustments:**

*Major revisions (we require the author to make these changes)*

Tables: table 3 has been divided into parts, a and b. It is important that tables are numbered in ascending numerical order: 1, 2 and 3. Please renumber the tables and any references to them in the text. Alternatively, please merge the tables into one larger table.

Tables: table 4 has been divided into parts, a and b. It is important that tables are numbered in ascending numerical order: 1, 2 and 3. Please renumber the tables and any references to them in the text. Alternatively, please merge the tables into one larger table.

*In the present version of the MS, we have taken care to merge both tables 3 and 4 into unique, undivided tables that fit on one sheet of paper.*

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.

*This has been done.*

If you have any questions or comments please do not hesitate to contact the corresponding author. Thank you for your continued interest in this manuscript. We look forward to hearing from you.

Sincerely yours,

Abdulrahman M. El-Sayed

For correspondence please contact:
Abdulrahman El-Sayed
Department of Epidemiology
Mailman School of Public Health
Columbia University
722 W. 168th Street
New York, NY 10032
Tel (US): +1 248 930 0522
e-mail: ame2145@columbia.edu